



### GOPEN ACCESS

**Citation:** Freedman AH, Schweizer RM, Ortega-Del Vecchyo D, Han E, Davis BW, Gronau I, et al. (2016) Demographically-Based Evaluation of Genomic Regions under Selection in Domestic Dogs. PLoS Genet 12(3): e1005851. doi:10.1371/journal. pgen.1005851

Editor: Matthew T Webster, Uppsala University, SWEDEN

Received: August 26, 2015

Accepted: January 18, 2016

Published: March 4, 2016

**Copyright:** This is an open access article, free of all copyright, and may be freely reproduced, distributed, transmitted, modified, built upon, or otherwise used by anyone for any lawful purpose. The work is made available under the <u>Creative Commons CC0</u> public domain dedication.

Data Availability Statement: Sequence data are available at <u>http://www.ncbi.nlm.nih.gov/bioproject/</u> PRJNA274504.

**Funding:** The authors received no specific funding for this work.

**Competing Interests:** CL, VT, and TTH were employed by Life Technologies during the time the research was conducted. Construction and sequencing of SOLiD Exact Call Chemistry libraries were also provided by Life Technologies. Life Technologies Corporation had no input into the **RESEARCH ARTICLE** 

## Demographically-Based Evaluation of Genomic Regions under Selection in Domestic Dogs

Adam H. Freedman<sup>1</sup>\*, Rena M. Schweizer<sup>1</sup>, Diego Ortega-Del Vecchyo<sup>1</sup>, Eunjung Han<sup>1</sup>, Brian W. Davis<sup>2</sup>, Ilan Gronau<sup>3</sup>, Pedro M. Silva<sup>4</sup>, Marco Galaverni<sup>5</sup>, Zhenxin Fan<sup>6</sup>, Peter Marx<sup>7</sup>, Belen Lorente-Galdos<sup>8</sup>, Oscar Ramirez<sup>8</sup>, Farhad Hormozdiari<sup>9</sup>, Can Alkan<sup>10</sup>, Carles Vilà<sup>11</sup>, Kevin Squire<sup>12</sup>, Eli Geffen<sup>13</sup>, Josip Kusak<sup>14</sup>, Adam R. Boyko<sup>15</sup>, Heidi G. Parker<sup>8</sup>, Clarence Lee<sup>16</sup>, Vasisht Tadigotla<sup>16</sup>, Adam Siepel<sup>17</sup>, Carlos D. Bustamante<sup>18</sup>, Timothy T. Harkins<sup>16</sup>, Stanley F. Nelson<sup>12</sup>, Tomas Marques-Bonet<sup>8,19</sup>, Elaine A. Ostrander<sup>2</sup>, Robert K. Wayne<sup>1</sup>, John Novembre<sup>1¤</sup>

1 Department of Ecology and Evolutionary Biology, University of California, Los Angeles, Los Angeles, California, United States of America, 2 National Human Genome Research Institute, National Institutes of Health, Bethesda, Maryland, United States of America, 3 Department of Biological Statistics and Computational Biology, Cornell University, Ithaca, New York, United States of America, 4 CIBIO-UP, University of Porto, Vairão, Portugal, 5 ISPRA, Ozzano dell'Emilia, Italy, 6 Key Laboratory of Bioresources and Ecoenvironment, Sichuan University, Chengdu, China, 7 Department of Measurement and Information Systems, Budapest University of Technology and Economics, Budapest, Hungary, 8 ICREA at Institut de Biologia Evolutiva (CSIC-Universitat Pompeu Fabra), Barcelona, Spain, 9 Department of Computer Science, University of California, Los Angeles, Los Angeles, California, United States of America, 10 Bilkent University, Ankara, Turkey, 11 Estación Biológia de Doñana EBD-CSIC, Sevilla, Spain, 12 Department of Human Genetics, University of California, Los Angeles, Los Angeles, California, United States of America, 13 Department of Zoology, Tel Aviv University, Tel Aviv, Israel, 14 Department of Biology, University of Zagreb, Zagreb, Croatia, 15 Department of Biomedical Sciences, Cornell University, Ithaca, New York, United States of America, 16 Life Technologies, Foster City, California, United States of America, 17 Simons Center for Quantitative Biology, Cold Spring Harbor Laboratory, Cold Spring Harbor, New York, United States of America, 18 Stanford School of Medicine, Stanford, California, United States of America, 19 Centro Nacional de Analisis Genomico (CNAG/PCB), Baldiri Reixach 4-8, Barcelona, Spain

¤ Current address: Department of Human Genetics, University of Chicago, Chicago, Illinois, United States of America

\* adamfreedman@fas.harvard.edu

## Abstract

Controlling for background demographic effects is important for accurately identifying loci that have recently undergone positive selection. To date, the effects of demography have not yet been explicitly considered when identifying loci under selection during dog domestication. To investigate positive selection on the dog lineage early in the domestication, we examined patterns of polymorphism in six canid genomes that were previously used to infer a demographic model of dog domestication. Using an inferred demographic model, we computed false discovery rates (FDR) and identified 349 outlier regions consistent with positive selection at a low FDR. The signals in the top 100 regions were frequently centered on candidate genes related to brain function and behavior, including *LHFPL3*, *CADM2*, *GRIK3*, *SH3GL2*, *MBP*, *PDE7B*, *NTAN1*, and *GLRA1*. These regions contained significant enrichments in behavioral ontology categories. The 3<sup>rd</sup> top hit, *CCRN4L*, plays a major role in lipid metabolism, that is supported by additional metabolism related candidates revealed in our

development of the study, article preparation or decision to publish. No other competing interests are declared.

PLOS

GENETICS

scan, including *SCP2D1* and *PDXC1*. Comparing our method to an empirical outlier approach that does not directly account for demography, we found only modest overlaps between the two methods, with 60% of empirical outliers having no overlap with our demography-based outlier detection approach. Demography-aware approaches have lower-rates of false discovery. Our top candidates for selection, in addition to expanding the set of neurobehavioral candidate genes, include genes related to lipid metabolism, suggesting a dietary target of selection that was important during the period when proto-dogs hunted and fed alongside hunter-gatherers.

#### Author Summary

Identification of the genomic regions under selection during dog domestication is extremely challenging because the demographic fluctuations associated with domestication can produce signals in polymorphism data that mimic those imposed by selective sweeps. We perform the first analysis of selection on the dog lineage that explicitly incorporates a demographic model, that by controlling for the rate of false discovery, more robustly identifies targets of selection. To do so, we conduct a selection scan using three wolf genomes representing the putative centers of dog domestication, two basal dog breeds (Basenji and Dingo), and a golden jackal as outgroup, for which we previously inferred a demographic model. We find that our demographically informed analyses filters out many signals that would be otherwise classified as putative selection signals under an empirical outlier approach. We identify 68 regions of the genome that have likely experienced positive selection. Besides identifying a number of new neurobehavioral candidate genes, our candidate regions contain genes related to lipid metabolism, including CCRN4L, which is centered in the 3<sup>rd</sup> ranked region. This suggests a previously unreported locus of dietary adaptation, potentially due to the change in diet composition as hunting efficiency increased when proto dogs began hunting alongside hunter-gatherers.

#### Introduction

Identifying regions of the genome that have undergone recent positive selection is central to understanding the causes of evolutionary diversification. Nevertheless, developing efficient and statistically robust methods for distinguishing genomic regions under selection from the neutral background expectation remains extremely challenging, particularly under complex, non-equilibrium demographic scenarios. The rapid rise in frequency of a new favorable allele typically leads to a reduced diversity in flanking regions as linked neutral polymorphism accompanies the adaptive substitution in a phenomenon known as genetic hitchhiking [1]. Many methods have been developed to detect such "selective sweep" signatures using genome-wide polymorphism data [2–4]. However, the distortions of the site-frequency spectrum (SFS) and/ or extended linkage-disequilibrium accompanying episodes of positive selection can be difficult to distinguish from that produced by neutral processes related to a specific demographic history. For example, coalescent trees produced by population bottlenecks or founder events may be indistinguishable from those generated by selection [5,6], and in general, bottlenecks can generate long haplotypes that mimic those observed in selective sweeps [7]. Furthermore, population subdivision can produce counterintuitive and confounding effects [8,9]. Consequently,

such demographic heterogeneity contributes to the low power and high false positive rates that can occur in genome-wide selection when using contemporary approaches [10-12].

The domestication of dogs from gray wolves is relevant to understanding the broader history of animal domestication and the genetic architecture of rapid phenotypic evolution [13,14]. As humans migrated out of Africa, they encountered gray wolves, which served as the founding stock for the domestic dog lineage. Archaeological remains [15–18], analyses of whole genome sequence data [19], and mitochondrial genomes of ancient and extant canid lineages [20] jointly support a pre-agricultural origin of dogs which was initiated by association with hunter-gatherers. During this initial interaction, selection for domestication traits was less intentionally directed by humans than it has been with the recent evolution of breed dogs, and instead, was predominantly an incidental by-product of human-wolf-prey interactions [21]. It is likely that dog domestication involved significant genetic changes in response to dietary and behavioral divergence from a wolf ancestor, and comparisons of brain-specific gene expression differences between dogs and wolves support the importance of the latter [22].

Identifying the targets of selection responsible for phenotypic divergence between wolves and dogs is hampered by the demographic complexity of the earliest phases of dog domestication, during which the ancestral dog lineage experienced at least one severe bottleneck and admixture with wolves occurred [19]. Such bottlenecks and admixture can bias selection scans that do not incorporate a demographic model, leading to false positive and negative signals, depending on the circumstances. Despite these potentially confounding effects, of the several studies investigating the genetic basis of phenotypic variation among recently formed breeds and early in domestication [23–28], none have formally modeled demography to generate a null, neutral expectation for patterns of variation.

Recently, we used coalescent-based analysis of whole genome sequence data from dogs and wolves to elucidate the complex demographic history underlying the domestication process. We estimated that domestication entailed a >16-fold reduction in effective population size  $(N_e)$  for dogs, and a weaker, 3-fold reduction in wolves that began shortly after the initial dog-wolf divergence [19]. By comparison to modern wolves, earlier studies inferred a weaker domestication bottleneck [28–30], but the ancestral wolf bottleneck had not been previously known, and thus our results showed a greater loss of variation because dogs descended from a more variable ancestral wolf population. We also found evidence for considerable post-divergence admixture, not only between dogs and wolves, but also between wolves and golden jackals, and between golden jackals and the dog-wolf ancestor [19]. Recent admixture between dogs and wolves [31], and admixture between wolves and coyotes [32] had been previously detected, but the extent to which admixture events may obscure dog origins has only recently been appreciated [18,21,33,34]. This combination of bottlenecks and admixture substantially complicates efforts to distinguish between neutral processes and natural selection.

Previous investigations of selection on the dog lineage have taken an approach sometimes referred to as an empirical outlier scan for selection in which putatively selected regions are identified as outliers falling above some arbitrary value [13,25,27,35]. While this approach will detect loci under intense selection, controlling the rate of expected false positives is difficult because the distribution of test statistics under a null demographic model are not taken into account. Similarly, other recent studies of selection in domestic [36,37] and wild populations [38] have not accounted for demographic complexity. One difficulty is that a complete demographic model for large genome studies requires a time-consuming investigation of alternative scenarios that is computationally intensive.

To investigate positive selection on the dog lineage early in the domestication process and prior to the recent diversification of breeds, we re-examine patterns of polymorphism at 10 million single-nucleotide variant sites using six previously sequenced canid genomes that were

used to infer a demographic model of dog domestication [19]. This sample included three wolves from Israel, Croatia, and China; two divergent dog breeds thought to be basal in the dog phylogeny, Basenji and Dingo; and a golden jackal [19]. Specifically, we use our previously inferred demographic model to calibrate a genome-wide scan for signatures of positive selection on the dog lineage and more confidently identify possible targets of recent positive selection while controlling for false positives. Although a recent genomic analysis of a wolf fossil has suggested a slower mutation rate for canids than used in our initial interpretation of our model [39], the raw parameter estimates from our model are independent of the mutation rate, i.e. our model explains the neutral distribution of polymorphism across our samples, regardless of the well-known uncertainty surrounding mutation rates. Finally, we contrast our findings with a demography-agnostic approach typical of previous studies. Our results expand the catalog of candidate neurobehavioral and dietary genes involved in domestication and provide candidates for future functional studies.

#### Results

By leveraging the dataset of Freedman et al. [19], we were able to compute three summary statistics that are sensitive to the effects of positive selection in sliding windows across the dog reference genome [40]. These three statistics are as follows: 1) the difference in nucleotide diversity between dogs and wolves ( $\Delta \pi$ ); 2) F<sub>ST</sub>; and 3) the difference between dogs and wolves in Tajima's D ( $\Delta$  TD). After filtering on genome and sample level features, we computed summary statistics for 195,998 100kb sliding windows incremented in 10kb steps. Considerable variation was observed in the distribution of the three summary statistics (S1 Fig), and in our composite-of-multiple-signals statistic, comprised of the product of 1-FDR for those statistics (CMS<sub>1-FDR</sub>, see <u>Methods and Materials; Fig 1</u>). We used coalescent simulations based upon our previously constructed demographic model [19] to evaluate these signals relative to a genomewide neutral expectation. Our model was inferred from a set of putatively neutral loci defined by a stringent set of filters with respect to features such as proximity to genic regions, degree of conservation, and the presence of segmental duplications. To estimate window-specific false





doi:10.1371/journal.pgen.1005851.g001

discovery rates (q-values [41] from our empirical data), we conducted 200,000 simulations of 100kb windows with parameters fixed to the mean posterior values inferred for our demographic model (S12 Table in [19]). After calculating our three summary statistics for the simulated windows, for each statistic in each observed window, we calculated a p-value as the probability of observing in the simulated windows a value equal to or greater than that in the observed window. We then used the Benjamini-Hochberg procedure to calculate the probability of false discovery given that p-value as a means to correct for multiple comparisons [42].

To contrast the FDR-based findings with those not explicitly incorporating demography, we also identified outliers using an empirical outlier method. In this approach, we identified outlier regions as those comprised of the top 1% of all 100kb regions based on the joint percentiles of the underlying summary statistics (see <u>Methods</u> for details) which is similar to that used in a previous assessment of selection in dogs [<u>35</u>]. We collapsed windows into regions using the same criteria as in our FDR-based method.

#### **Regions under selection**

Comparison of our observed data to summary statistics observed in 200,000 simulations of 100kb windows under our previously inferred demographic model indicated a general overdispersion of empirical windows relative to simulated ones. While some of this over-dispersion may be due to heterogeneity in genomic features (e.g. mutation rate) and the collective impact of various evolutionary processes, there is a clear excess of extreme values falling in the right-hand tails, outside the distribution of neutrally evolving windows, and consistent with the action of positive selection (Fig 2). Employing a false discovery rate (FDR) of 0.01 for  $\Delta \pi$ , F<sub>ST</sub>, and  $\Delta$  TD statistics, we identified 353, 827, and 982 windows, respectively, bearing signals consistent with positive selection (Table 1), for a total of 2081 unique windows. As an alternative approach, we repeated the procedure using null simulations with parameters drawn from the joint posterior distribution rather than fixing them at their mean posterior values (see Methods). The distributions of summary statistics were similar under both approaches (S2 Fig, Pearson correlations between FDR estimates between each approach >0.999, P < 2.2 × 10<sup>-16</sup>, 2558 unique windows identified). To be conservative, our subsequent analyses focus on the more limited set of 2081 windows found using both approaches.

After joining significant windows that were  $\leq$  200kb apart, both within and across statistics, 349 regions remained in total (<u>S1 Table</u>). These regions overlapped only partially with those identified in previous studies of selection in dogs. Specifically, 53 regions from previous studies were recovered using our approach, and additionally, we detected 296 novel regions.

With the 1% threshold, the empirical outlier approach identified 309 outlier regions. The overlap between the FDR-based and empirical outlier methods was low: 59% of the loci based on the FDR-based approach had no overlap with those from the empirical outlier method and 60% of empirical outliers had no overlap with the FDR-based approach (S3 Fig). Two patterns help to explain the low degree of overlap between the methods. First, looking at each summary statistic separately, the vast majority of windows in the top 1% of the empirical distribution have an FDR that would not pass our threshold of 0.01 (S4 Fig). A similar pattern is observed with the joint percentile statistics in that the vast majority of windows with an empirical joint percentile in the top 1% have high FDR for individual statistics (S5 Fig, red points), and in many cases more than one statistic. In both cases such outlier windows would be excluded using our FDR-based method. These results suggest that, in the absence of a baseline to assess if signals are consistent with neutral evolution, more than half of outliers in the empirical approach are not supported by an FDR-based approach, and many may actually be false positives. Furthermore, at the gene-level, the FDR and empirical outlier methods identify



Fig 2. Distributions of observed values for selection scan statistics and those computed from neutral coalescent simulations based up the inferred demographic history [19] for (A)  $\Delta \pi$ , (B) F<sub>ST</sub>, and (C)  $\Delta$  Tajima's D. Dashed lines indicate threshold values for FDR  $\leq 0.01$ .

doi:10.1371/journal.pgen.1005851.g002

substantially different sets of genes, with 64% of genes identified in empirical outliers falling outside of FDR-identified regions. This suggests that inferences without demography might lead to mistaken functional interpretation of putative selection signals and gene ontology enrichments (S6 Fig).

Statistic	FDR	Windows <fdr< th=""><th>Minimum value <math>\leq</math> FDR threshold</th><th>P at FDR</th></fdr<>	Minimum value $\leq$ FDR threshold	P at FDR
Δπ	0.05	519	3.127	1.25 × 10 <sup>-2</sup>
	0.01	353	3.332	$1.0 \times 10^{-5}$
F <sub>ST</sub>	0.05	1495	0.481	$3.8 \times 10^{-4}$
	0.01	827	0.544	$4.0 \times 10^{-5}$
Δ Tajima's D	0.05	2329	1.466	$6.0 \times 10^{-4}$
	0.01	982	1.763	$5.0 \times 10^{-5}$

#### Table 1. FDR threshold values and window counts for selection scan statistics.

doi:10.1371/journal.pgen.1005851.t001

To rank the putative regions under selection we used a composite-of-multiple signals approach [43]. Specifically, we computed window-specific probabilities of false discovery (i.e. a false inference of deviation from neutrality) for our three summary statistics, and then computed the product of 1-FDR across those statistics to obtain a quantity we label  $CMS_{1-FDR}$ . As the three summary statistics are not independent within windows, this product does not scale exactly with the weight of evidence for positive selection. Nevertheless, larger values of this statistic should indicate regions that are less likely to have been evolving neutrally. We used the maximum  $CMS_{1-FDR}$  statistic observed for any outlier window to rank windows and to localize the selection signal within each region (Fig.3). This statistic localizes the selection signal within outlier regions more tightly than computing a joint empirical percentile statistic (S7 Fig) which does not explicitly incorporate the probability of observing any of the constituent statistics under neutrality. When describing specific candidate genes likely under selection, we employ an additional filter in order to minimize false positives, by considering only genes within the top 100 regions.

The joint distribution of summary statistics, joint percentile, and CMS<sub>1-FDR</sub> for 100kb window highlights the potential problems of not explicitly incorporating demography into selection scan for our set of genomes. To visualize these problems, we classify 100kb windows into four categories. The first category consists of those windows with both a low  $CMS_{1-FDR}$  statistic and high FDR for all three summary statistics, falling completely within neutral expectations ("low CMS, high FDR" in Fig 4A and 4C). It is possible for a window to have FDR $\geq$ 0.01 for all three statistics, but still have low enough FDR such that  $CMS_{1-FDR}$  is comparable to that observed in outlier regions. We distinguish high  $CMS_{1-FDR}$  windows as those with a value for this statistic greater or equal to that observed in the top 100 ranked regions (i.e. the minimum across those 100 regions of the maximum value observed within a region). Thus, the second category consists of sites with FDR ≥0.01 across all three summary statistics, but CMS<sub>1-FDR</sub> above this threshold ("high CMS, high FDR" in Fig 4A and 4C). In some cases, windows have  $FDR \le 0.01$  for at least one summary statistic but there is at least one statistic with high FDR, such that they are classified as deviating from neutrality while having relatively low CMS<sub>1-FDR</sub>, beneath the threshold defined above ("low CMS, low FDR in Fig 4A and 4C). Finally, there are windows that have consistently low FDR across statistics such that CMS<sub>1-FDR</sub> is high, owing to consistent signals of selection across statistics ("high CMS, low FDR in Fig 4A and 4C).

Based upon this classification of windows, we can distinguish different types of evidence for positive selection (Fig 4A and 4C). In contrast, many of the windows identified by the joint percentile method have high FDR for all three statistics (contrast blue points in Fig 4A and 4C with red points in Fig 4B and 4D), or have high enough FDR for some statistics such that  $CMS_{1-FDR}$  is low (contrast orange points in Fig 4A and 4C with red points in Fig 4B and 4D). However, by restricting our analysis to the top 100 windows we exclude regions that would be



Fig 3. Z-transformed selection scan statistics, CMS<sub>1-FDR</sub>, and gene annotations within the (A) top ranked, (B) 3<sup>rd</sup> ranked, (C) 4<sup>th</sup> ranked, and (D) 5<sup>th</sup> ranked candidate regions for positive selection on the dog lineage.

doi:10.1371/journal.pgen.1005851.g003

flagged by such low CMS<sub>1-FDR</sub> windows that have very low support across all three summary statistics.

Key functional changes that derive from selection during the domestication process involve brain function and behavior [27,28,35], diet and metabolism [27], and pigmentation [44]. Consequently, we focus our discussion of the results on genes in regions showing evidence of a selective sweep with the FDR-based approach that are potentially relevant to these phenotypes. We only report genes that either overlap with the peak of the  $CMS_{1-FDR}$  statistic within an outlier region, or those that appear most proximate to that peak signal. As a further filter, we evaluated diversity patterns in 500kb intervals surrounding our top 100 outlier regions in a broader panel of 12 diverse breed dogs sequenced to approximately 40x mean coverage (SRA PRJNA288568). These sequence data include the dingo and basenji used in Freedman et al. [19] and genotypes were called for these data in a manner analogous to [19]. Based on these

# 



Fig 4. Biplots of summary statistics for 100kb sliding windows classified by their (A, C) CMS<sub>1-FDR</sub> and (B, D) joint percentile.  $CMS_{1-FDR}$  is classified according to whether it is  $\geq$  the minimum value observed in the top 100 regions for the maximum of  $CMS_{1-FDR}$  comprising the region (i.e. "high CMS"), and whether at least one summary statistic has an FDR  $\leq$  0.01 (i.e. low FDR). Thus, windows can be classified as "low CMS, high FDR", "high CMS, high FDR", "low CMS, low FDR", and "high CMS, low FDR." The first two categories are consistent with neutral expectations, the third is characterized by very weak evidence for selection, and the last category includes those windows with the strongest evidence for selection. For more details on these categories, see **Regions under selection** in <u>Results</u>.

doi:10.1371/journal.pgen.1005851.g004

data, we excluded from further consideration any of the top 100 outlier regions where diversity in the 12-breed panel was greater or equal to that in adjacent non-outlier regions, or where the outlier region was centered on a localized reduction in diversity comparable to those seen in adjacent non-outlier intervals. This confirming data resulted in a reduced set of 68 regions. The filtered set of regions overlapped with only 21 previously identified candidate regions, and contained 47 novel regions ( $\underline{Fig 5}$  and  $\underline{S8 Fig}$ ).

In some cases, for any given outlier region, more than one gene may meet our criteria outlined above, such that highlighting particular genes will be *ad hoc*. Furthermore, it is possible that focusing on particular genes may exclude un-annotated regulatory elements that alter expression of downstream genes more distant from the statistical signal of selection. These

Region		This S	Study	~ ~ ~	Can	Map I	Data	Ot	her s	Studies		Genes
19:15 40 15 79	Δπ	Fst	ΔTD	CMS		Fst	Joint	1	2	3	4	
10.15.49-15.70	4.03	0.03	1.74	0.97	0.76	0.7	99.20	94.0				
18:0.57-1.02	4.36	0.67	1.62	0.97							28	
19:3.55–3.72	3.16	0.51	2.09	0.93	0.58	0.83	99.22	93.1				CCRN4L
31:1.95-2.14	2.9	0.65	2.17	0.92	0.54	0.4	94.22					CADM2
12:59.02-59.17	3.93	0.64	1.32	0.92	-1.06	0.01	0.57					RPL3;ACA64
10:4.04-4.22	2.42	0.53	2.21	0.9	0.3	0.61	96.02				18	SNORA31
11:37.55-37.76	3.18	0.69	1.62	0.89	0.03	0.47	79.33					SH3GL2
24:4.37-4.59	2.17	0.59	1.91	0.89	0.12	0.12	<mark>48.36</mark>					DTD1
24:23.25-23.48	8.91	0.69	1.4	0.89					0.7	0.64		ASIP;AHCY;unknown
2:21.9-22.16	1.87	0.93	2.14	0.88	0.18	0.19	68.71					FAM107B
32:0.64-0.78	4.09	0.43	2.66	0.88								ART3;NUP54;unknown;SCARB2
13:46.85-47.02	3.54	0.42	2	0.87	0.19	0.54	92.4	92.5				2 2 2 2 2 2 2 2 2 2
13:53.99–54.13	3.71	0.43	1.65	0.87	0.05	0.3	75.43					unknown
1:2.93-3.19	2.75	0.75	2.13	0.86	-0.04	0.13	<mark>40.36</mark>				1	MBP;ZNF236
1:28.36-28.52	2.83	0.6	1.52	0.85	0.1	0.51	87.23	93.3				PDE7B
2:46.35-46.53	3.19	0.43	2.4	0.85	<mark>-1.39</mark>	0.34	3.67					
6:28.39-28.49	2.1	0.45	1.79	0.83	0.24	0.52	93.8					unknown;PDXDC1;NTAN1;RRN3
4:57.36-57.48	2.49	0.45	1.87	0.82	0.16	0.62	91.71					GLRA1
37:16.14-16.27	1.9	0.46	2.5	0.82	0.47	0.67	98.18	91.5				PLEKHM3
4:60.02-60.27	3.71	0.6	1.01	0.81	-0.14	0.52	54.44					
15:9.83-9.93	3.48	0.47	1.12	0.81	<mark>-0.36</mark>	0.4	<mark>31.01</mark>					EPS15;TTC39A
8:71.21-71.31	3.11	0.4	1.79	0.8								MARK3;U2;CKB;
4:61.87-62.1	4.16	0.43	1.89	0.8	0.04	0.21	64.58					unknown
16:39.84–39.99	3.44	0.39	2.15	0.8	-0.11	0.07	19.51					
4:4.17-4.34	2.54	0.42	2.04	0.8	-1.29	0.06	1.34					LYST;unknown;GNG4

Fig 5. Top 25 outlier regions identified using the FDR-based methodology using  $\Delta$ out  $F_{ST}$ ,  $\Delta$  Tajima's D and validated with the 12-breed dog diversity panel (see text), with regions ranked according their respective maximum CMS<sub>1-FDR</sub> statistic. Columns within "This study" are based on the sequencing data generated here, while those under "CanMap" are computed from a ~48k SNP data set for a large set of wolves and ancient/basal dog breeds [35]. Heat map colors reflect upper percentiles of the calculated metrics, with warmer colors indicating higher percentiles. Overlaps with previous studies: 1, vonHoldt et al. 2010 [35]; 2, Vaysse et al. (2011) [25]; 3, Boyko et al. (2010) [23]; and Axelsson et al. (2013), [27]; with numbers indicating the joint percentile,  $F_{ST}$ ,  $F_{ST}$  and region id, respectively for each study.

doi:10.1371/journal.pgen.1005851.g005

caveats aside, we emphasize that our goal is to provide an updated list of candidate genes that can be used as a resource on which to base future investigations and functional assays, rather than to make absolute claims about the importance of any one gene to the domestication process. On a region-by-region basis, we document the extent to which the reported gene is the only one in the putative sweep region or whether it is the gene closest to the peak of the  $CMS_{1-FDR}$  statistic. Fig 5 and S8 Fig, provide a summary of the top regions we present given these considerations.

#### Brain function/behavior genes

Eight of the top 20 candidate regions contain genes that have been implicated in neurological functions in other mammalian species. Our top region is centered on *LHFPL3*, a member of the lipoma *HMGIC* fusion partner family (Fig 3A). Mutations in *LHFPL3* have been detected in malignant glioma patients [45] and associated with autism risk [46]. *CADM2* is located within the 4th most extreme outlier region (Fig 3D) and is a synaptic cell adhesion molecule whose flanking regions show reduced homozygosity in autism patients [47]. *GRIK3* is the only gene within the 6<sup>th</sup> region, and overlaps with the peak in the CMS<sub>1-FDR</sub> signal. It is a glutamate receptor that has been associated with personality traits such as harm avoidance [48], schizophrenia and bipolar disorder [49], and was a neurobehavioral candidate gene in a selection scan of domestic cattle [36]. One cautionary note is that within this region our filters exclude large regions immediately adjacent to it, which raises the possibility that local genomic features might influence the quality of genotype calls.

SH3GL2 is the only gene proximate to the peak in the  $CMS_{1-FDR}$  within the 8th ranked region and affects synaptic vesicle formation [50]. The peak signal in the 16<sup>th</sup> ranked region is closest to *MBP*, a major constituent of the myelin sheath of oligodendrocytes and Schwann cells, and shown to be involved in schizophrenia [51]. *PDE7B*, which is the only gene overlapping the 17<sup>th</sup> ranked region, is highly expressed in the brain and is involved in striatal functions related to dopaminergic pathways [52]. Inactivation of *NTAN1* (19<sup>th</sup> region) in mice impairs spatial memory and leads to compensatory gains in non-spatial learning [53,54]. However, a RNA polymerase I-specific transcription initiation factor (*RRN3*) and *PDXDC1*, a gene with carboxylase activity associated with diverse phenotypes including renal carcinoma [55] and sensorineural hearing loss [56] were also either proximate to or overlapping the peak in CMS<sub>1-FDR</sub> signal. *GLRA1* (the only gene in the 20<sup>th</sup> region) mediates postsynaptic inhibition in the central nervous system, and mutations have been associated with startle disease [57]. For information on the remaining candidate genes with potential connections to behavior see <u>S1 Text</u>.

#### Diet/lipid metabolism genes

In our  $3^{rd}$  top outlier, the putative selection signature is most strongly peaked on *CCRN4L* (Fig <u>3B</u>). *CCRN4L* (also known as Nocturnin) is expressed in a circadian fashion and studies in mice indicate that *CCRN4L* activates *PPAR-* $\gamma$ , a gene that promotes bone adipogensis as opposed to osteoblast formation and that harbors a known diabetes risk variant in humans [58]. It also is known to regulate the expression of genes involved in lipogenesis and fatty acid binding, and knock-out mice are remarkable in being resistant to diet-induced obesity [58–61]. *CCRN4L* also suppresses *IGF1*, a well-known activator of bone growth [61] that underlies size variation amongst dog breeds [62,63]. The direction of these pleiotropic effects of *CCR4NL* implies a gain-of-function mutation would promote adipocyte formation, alter lipid metabolism, and suppress bone-growth.

Within our 9<sup>th</sup> region, a second peak in CMS<sub>1-FDR</sub> is centered on *SCP2D1*, a paralog of sterol carrier protein 2 (*SCP2*), which is highly expressed in genes involved in lipid metabolism, thought to function as an intracellular lipid transfer protein, and for which mice knockouts present altered lipid metabolism [64]. *PDXC1*, found within the 19<sup>th</sup> region in addition to *NTAN1* (see above), is associated with plasma phospholipid concentrations and is functionally connected to the glycerophospholipid and sphingolipid pathways [65]. For information on additional candidate genes see <u>S1 Text</u>.

#### Pigmentation candidate genes

The 10<sup>th</sup> top region was centered on agouti signaling protein (ASIP), a well-known gene influencing pigmentation in mammals [66,67], that has a lesser known role in inhibiting lipolysis [68]. More recently, evidence is emerging that variation at *ASIP* can influence social behavior, most likely through its antagonistic effects on melanocortin receptors or  $\alpha$ -melanocortin stimulating hormone [69,70]. Other than a small, predicted gene of unknown function, *LYST* is the only gene in the 30<sup>th</sup> region. *LYST* not only overlaps the peak CMS<sub>1-FDR</sub> signal, it overlaps the majority of the region as well. *LYST* has been associated with eye color variation in humans [71], and mutations can produce lighter skin and hair pigmentation [72].

#### Characterizing dog-specific mutations in outliers

We found 8883 sites (2226 in outlier regions) containing dog-specific mutations that were at high allele frequency in the 12-breed panel (S9 Fig). Sites fixed between the dog and wolves we sequenced were enriched in outliers with respect to functional class relative to other genomic regions ( $\chi^2 = 23.06$ , df = 9, P =  $6.1 \times 10^{-3}$ ). The relative abundance of fixed differences in regions within one kb upstream of the transcriptional start site was twice that of the neutral background. Even so, there were only 12 upstream dog-specific mutations in outlier regions (S9 Fig), representing only 0.5% of all fixed sites in outlier regions. In contrast, the majority of dog-fixed sites fall within introns (29.2%) and putative intergenic (68.0%) regions. Only eight non-synonymous fixed sites were observed in outlier regions, and only five within regions that showed reduced diversity in the12-breed panel. Ensembl's Variant Effect Predictor tool predicted that, for the transcript annotation displaying the maximum effect, all five variants were mutations of moderate effect. Associated SIFT predictions were as follows: SLK, in 115th ranked region, low-confidence deleterious; two mutations in ACSBG2, 135<sup>th</sup> ranked region, deleterious and tolerated, respectively; NOL8 (uncharacterized protein), 292.5th ranked region, tolerated; ZNF585B, 292.5<sup>th</sup> ranked region, tolerated. The one high confidence deleterious prediction based upon SIFT is in ACSBG2, which encodes a protein that is testis and brain-specific, and may play a role in spermatogenesis [73]. Nevertheless, the low frequency of dogspecific non-synonymous fixed sites and their occurrence within relatively low ranked outlier windows suggest coding mutations have been less important in the phenotypic divergence between dogs and wolves.

#### Enrichment analyses

For enrichment analyses, we focused on the top 100 regions ranked by  $CMS_{1-FDR}$ , minus those that did not also show reduced diversity in the 12-breed data set. We further filtered the gene set by only considering all genes that fell within 25kb of the peak in  $CMS_{1-FDR}$  within those regions. Based upon our requirement that FDR was  $\leq 10\%$ , we identified three categories that showed evidence of enrichment in the outlier regions. Notably, we found enrichments for behavior, locomotory behavior, and adult behavior (Table 2). However, after correction for multiple tests, none of these categories was significant. While it has been suggested that family-

Table 2. Enrichment categories discovered from the top 100 regions within 25kb of peak in joint statistic signal, excluding regions that fail to show reduced diversity in the 12-breed data set and categories with FDR >10%. Input and background total number of genes are 50 and 13,528, respectively.

Category	Ρ	P-corrected <sup>a</sup>	FDR (%)	Background in Term	Fold Enrichment <sup>b</sup>	Genes
Biological process: behavior	0.0014	0.5877	2.1	469	4.6	DOCK2, GLRA1, LYST, ABAT, NTAN1, MBD2, ASIP, CXCL10
Biological process: locomotory behavior	0.0030	0.6163	4.4	274	5.9	DOCK2, GLRA1, LYST, ABAT, NTAN1, CXCL10
Biological process: adult behavior	0.0037	0.5421	5.4	86	12.6	GLRA1, ABAT, NTAN1, ASIP

<sup>a</sup> Benjamini-corrected P-value.

<sup>b</sup> (number of enriched genes in GO term/50)/(background genes in GO term/13528).

doi:10.1371/journal.pgen.1005851.t002

wise control of Type I errors is overly conservative for enrichment analyses [74], we consider our enrichment findings tentative, albeit consistent with the frequent appearance of brain function/behavior genes in our top hit regions.

#### Discussion

Extreme population bottlenecks are a hallmark of domestication events, and, in particular, demographic fluctuations and frequent admixture are regarded as important features of the evolutionary history of dogs [18,21,33]. We present the first effort to control for potential confounding effects of bottlenecks when inferring positive selection on regions of the dog genome, using a robust demographic model constructed from the same set of samples used to perform selection scans.

Two categories of genes continually emerged in the top half of our candidate regions list: those influencing behavior, neuropsychiatric disorders and brain function, and genes related to metabolism, in particular lipid metabolism. Genes associated with brain function and behavior are expected, given the dramatic shift from wild to domestic existence. However, genes related to fat metabolism are more surprising, and complement previous evidence for dietary adaptation occurring during domestication, particularly for increased starch metabolism [27]. Our 3<sup>rd</sup> ranked region is nocturnin (*CCRN4L*). Evolution at this locus and at other metabolism genes (e.g. *ADRB2*, *DIP2C*, *PLCXD3*) may have facilitated shifts in lipid content of early domestic dog diets as they scavenged more on carcasses left behind by early humans. In fact, as incipient dogs and early humans began hunting together, prey capture rates may have increased relative to wild wolves and with it, the amount of lipid consumed by the assisting protodogs [75,76]. Unique dietary selection pressure may have resulted both from the amount consumed and the shifting composition of tissues that were available to protodogs after humans removed the most desirable parts of the carcass.

In addition to genes that may influence behavior and lipid metabolism, our selection scan also identified regions containing genes known to influence pigmentation. The effects of domestication on pigmentation are likely complex, potentially involving a combination of relaxed selection for crypsis, as well as positive selection for particular coat patterns [21]. The classic experiment selecting for tameness in foxes produced piebald and spotted coat color patterns after only 10 generations [44], suggesting that selection on pigmentation might not be direct but a by-product of selection on behavioral traits. While one investigation found no genetic correlation between coat coloration and tameness in rats [77], it is possible that in other species these two traits might be functionally coupled. As some of the pigmentation genes in our selection scans influence additional traits, the selection signals we detected may be

produced by direct selection on early dog pigmentation phenotypes—the nature of which is not yet clear—or via other traits influenced by such putative pigmentation genes (e.g. the K locus, Anderson et al. [31]).

The general trend of reduced fear/aggression in domesticated species raises the possibility that this behavioral shift may have involved selection on the same set of genes in different domesticated species [78]. Although a comprehensive analysis of neurobehavioral candidates across species is beyond the scope of this paper, there are some notably parallelisms. In cats, both glutamate receptor (*GRIA1*) and protocadherin (*PCDHA1*, *PCDH4B*) genes show evidence of positive selection [79]. Similarly, in our top 10 candidate regions, we observe a glutamate receptor (*GRIK3*) which was also identified in a selection scan in cattle [36]. In contrast, top neurobehavioral candidate genes in rats did not overlap with our candidate gene set [80]. These comparisons suggest that positive selection during domestication may act on particular pathways, such as glutamate receptors, but not necessarily the same genes within those pathways.

The simultaneous appearance of multiple traits during domestication, labeled the "domestication syndrome," raises fundamental questions concerning the genetic architecture of trait correlations. A recent, as of yet untested, hypothesis is that such correlations are functionally connected early in development during the processes of stem cell proliferation, differentiation, and migration [81]. Additionally, our results also suggest that pleiotropy may play a role in generating trait complexes observed in domestication species. For example, *CCRN4L* (our 3<sup>rd</sup> top hit) directly influences lipid metabolism, but may indirectly reduce body size through suppressive effects on the well-established growth regulator *IGF1*. As an additional example, variation at agouti can influence lipid metabolism and behavior as well as pigmentation. While these examples may point to a mechanism facilitating the domestication syndrome, validation of potential pleiotropic effects among the candidate genes within our outlier regions will require analyses of tissue-specific expression and focused functional studies.

Our candidate regions contain a number of potential targets of selection not observed in recent selection scans, and only overlap to a small degree with regions detected by previous studies on dogs (Fig 5 and S8 Fig). While the lack of reproducibility of candidate regions among studies has raised questions concerning their general utility [82], we attribute discordance with prior studies to several factors. First, we employed a two-level filtering scheme on genotypes that included excluding genotypes intersecting with genome-level features such as copy number variants, where incorrect read mappings will distort allele frequency estimates and summary statistics that rely on those estimates. For example, the filters we used exclude the copy number variable amylase gene that had been reported previously as a crucial target of selection during dog domestication [27]. In that regard, one caveat of our study is that we only will detect adaptations based on copy number variants or structural variation through their effects on linked single nucleotide variation.

Second, methods that explicitly incorporate demographic information will likely produce different results from those that do not. This is perhaps most clearly demonstrated by the lack of overlaps between our FDR-based and demography-free joint percentile method (Fig 5), the latter being characteristic of "empirical" approaches which can potentially miss key targets of selection and falsely identify others [82].

Third, the set of genomes evaluated can have an effect on which regions are identified. From our previous demographic analysis [19], we determined that admixture between dogs and wolves is geographically structured such that the probability of gene flow is higher for wolf and dog lineages that are geographically proximate. Thus, biased sampling of dogs towards particular breeds may confound selection scan results in dogs by revealing specific features of regional dog breeds and wolves. Interestingly, our candidate regions did show some overlaps with an

empirical outlier approach using SNP chip genotyping data when we restricted our sampling to so called "ancient" breeds (Fig 5 and S8 Fig). This overlap presumably occurs because our genomes and the ancient breed panel both retain patterns of polymorphism typical of the earliest dogs.

Our model-based assessment of FDR represents the first effort to account for demography in understanding positive selection during dog domestication. While this should reduce false positives among our candidate regions, a few caveats are necessary. First, our analysis is based on a small number of genomes. As a result, our approach should provide sufficient power for sweeps that have been strong, but partial sweeps that have led to less dramatic changes in dog allele frequencies will likely be missed. The challenge for future work is to expand the number of genome sequences analyzed while grounding selection scans with a demographic model that considers the intricacies of population dynamics and inter-lineage admixture. In particular, modeling demography for dozens to hundreds of lineages will pose a substantial inferential and computational challenge. A second caveat is that partial sweeps and soft sweeps may be difficult to detect with the summary statistics used here. It has been recently suggested that soft sweeps are the dominant mode of adaptation in wild populations [83] but there is still considerable uncertainty [84]. A third caveat is that, despite employing a stringent set of filters on both genome and sample level features, we cannot rule out the possibility that clusters of genotyping errors may have occurred in samples from either the dog or wolf lineage, such that some outlier regions may be false positives. Finally, while our use of overlapping, sliding windows allows us to localize the peak signal in outlier regions, it does raise the issue of non-independence and how it affects our approach to controlling for false discovery rate. Previous work indicates that the Benjamini-Hochberg FDR correction should be robust to certain kinds of dependence structure if tests meet the "positive regression dependency on each from a subset" (PRDS) criterion [85]. Furthermore, evidence has been presented that linkage mapping and associated tests fulfill PRDS [86], and the dependency among statistics at SNPs in such cases should very similar to that observed among statistics computed over windows across the genome. Nevertheless, should some features of our genome scans violate PRDS, it would mean that our estimates of FDR would be slightly less conservative (although certainly more so than empirical outlier approaches). We consider this an area worthy of future investigation.

Regardless of which mode is dominant, future work will likely uncover additional loci that have undergone positive selection in canids. In particular, future analyses using a larger set of dog and wolf genomes should provide power for assessing potential changes in adaptive substitutions occurring in multiple canine lineages, particularly if a neutral expectation can be calculated using a demographic model inferred for this larger sample. Despite these concerns, our model-based approach identifies a substantial number of new behavioral, metabolic, and pigmentation candidate genes that may contribute to the remarkable success of the oldest domesticated species and the only large carnivore adapted to life with humans.

#### Methods

#### Genome sequencing, sequence alignment and genotyping

All sequence alignment, genotyping, and quality-filtering methods were described previously [19]. Genotypes for all six canid genomes in that study were benchmarked against high quality genotypes from the Illumina CanineHD BeadChip, and showed a high degree of concordance with the chip data (e.g., 99.4% – 99.9% of heterozygous genotypes are confirmed by the CanineHD BeadChip). Sequence data are available at <a href="http://www.ncbi.nlm.nih.gov/bioproject/">http://www.ncbi.nlm.nih.gov/bioproject/</a> PRJNA274504. Vcf files can be obtained via the Dryad data repository at doi:10.5061/dryad. <a href="https://www.scbi.nlm.nih.gov/bioproject/">sk3p7</a>.

We chose lineages to sequence with the goal of elucidating the timing, demographic context, and geographic origins of dogs. We selected the Basenji and Dingo for sequencing, as they represent two divergent breeds basal on the dog phylogeny [35]. We also utilized the Boxer reference genome as an additional haploid chromosome set. The Chinese, Croatian, and Israeli wolves represent lineages sampled geographically from the three regions from which dogs were previously hypothesized to have originated (East Asia, Europe, and Middle East, respectively). The golden jackal was chosen as an outgroup. This sampling strategy is also informative for understanding selection early in the dog lineage, as it captures the range of variation found in both dogs and wolves, thus minimizing the confusion of selection signals from later, lineage-specific effects, such as might occur were we to bias sampling towards modern breeds of European origin. All genotypes initially generated in CanFam 3.0 reference genome coordinates by Freedman et al. [19] were converted to the most current version, CanFam 3.1.

#### Mapping regions with recent selective sweeps

Summary statistics. Demographic factors, such as population expansions, bottlenecks, or population structure, are confounders that distorts expected signatures of recent positive selection [9,11,12,87]. Since most domesticated species experience a population bottleneck [88], and a frequent mode of selection during domestication is selection from standing variation [40,89], detection of positive selection during domestication is difficult. To detect selective sweeps on the dog lineage during domestication, we selected three statistics that have been shown to have the highest power to detect selection under these conditions [89]:  $F_{ST}$  [40],  $\Delta \pi$ [40], and  $\Delta TD$  [40,90]. We used a sliding window approach in which we divided the reference genome into overlapping windows of size 100kb with 10kb increments. For each 100kb-window, we computed summary statistics using only sites that passed the genome and samplelevel filters [19]. We considered the boxer reference haplotype when we compute statistics within the dog sample or between the dog and wolf sample. Because our analysis included a mixture of haploid (boxer reference) and diploid samples, we calculated F<sub>ST</sub> from estimates of nucleotide diversity, i.e.  $(\pi_{between} - \pi_{within})/\pi_{between}$ , where nucleotide diversities are average per site estimates calculated across all pass filter sites within a 100kb window. We computed  $\Delta\pi$  as  $\pi_{\text{wolf}}/\pi_{\text{dog}}$  in each window, and report values on a log-scale (i.e.  $\log(\Delta\pi)$ ) and  $\Delta$ TD is computed as the difference in Tajima's D between the wolf and dog sequences. In cases where  $\pi_{dog}$ was zero, we added a small fractional increment so that  $\Delta \pi$  would still be computable. In cases where no segregating sites within dogs exist in a window, we did not calculate  $\Delta$ TD.

**Window filtering.** We obtained 195,998 sliding windows of size 100kb with 10kb increments genome-wide. We then discarded any windows in which the number of fully observed sites is less than 30kb, because it is more likely that those windows are within or close to repeat/CNV regions or regions of poor sequencing quality.

Identifying outlier regions. In order to minimize the confounding of neutral, demographic signals with those produced by positive selection, we developed an approach to control for the false positive rate with simulations. Using the posterior mean parameter estimates from the best-fitting model (Fig 5A in [19]), we first simulated 200,000 100 kb windows using the program ms [91] under the demographic model that we previously inferred from the same seven genomes (including boxer) analyzed here [19], and computed  $\Delta \pi$ , F<sub>ST</sub>, and  $\Delta$ TD for those windows. Next, for each window in our observed data, we computed the proportion of simulated windows with a test statistic  $\geq$  that in an observed window, i.e. a p-value defined as the probability of the observed data under neutrality. We then computed FDR using the Benjamini-Hochberg procedure for our set of windows based upon our empirical data. For each test statistic, we retained windows with FDR $\leq$ 0.01. We considered the effects of simulation strategy on FDR calculations and outlier window identification by drawing 1000 samples from the joint posterior distribution of our best-fitting demographic model and simulating each of these samples 200 times for a total of 200,000 100kb windows. As the sliding window approach means that windows are often clustered across the genome (Fig 1), we collapsed windows across statistics into outlier regions if they were within 200kb of each other. As a heuristic to rank outlier regions, we took a "composite of multiple signals" (CMS) approach, similar to that employed in a recent selection scan in humans [ $\underline{43}$ ]. Specifically, we computed CMS<sub>1-FDR</sub> as (1-FDR<sub> $\Delta\pi$ </sub>)\* (1-FDR<sub>FST</sub>)\*(1-FDR<sub> $\DeltaTD$ </sub>), and ranked regions according to the maximum CMS<sub>1-FDR</sub> observed in any of the 100kb windows comprising a region. We note that given  $\Delta$  Tajima's D cannot be calculated for windows with no polymorphism, we have excluded windows with strong signals in F<sub>ST</sub> and  $\Delta\pi$  that are potentially of interest (orange points on far right of Fig 4A).

To contrast our demography-informed approach of controlling for FDR with neutral simulations, we examined overlap of outliers based upon this approach with a fundamentally different, "demography-free" approach that examines an arbitrary percentage of a top set of outliers, similar to many genome-wide selection scans that don't explicitly include demography. In this approach, for each summary statistic (F<sub>ST</sub>,  $\Delta\pi$  and  $\Delta$ TD), we computed empirical percentiles by ranking each window by the summary statistic in question and transforming the ranks to percentiles (%  $F_{ST}$ , %  $\Delta\pi$  and %  $\Delta$ TD). We then calculated a "joint" empirical percentile by computing the product of the empirical percentiles obtained for the three summary statistics in each window [(% Product) = (%  $F_{ST}$ ) \* (%  $\Delta\pi$ ) \* (%  $\Delta$ TD)] and then ranking each window by the products (% Product) and transforming the ranks to percentiles (% Joint). In order to draw Manhattan plots, we transformed the joint empirical percentiles defined for each window into joint empirical p-values. Joint empirical p-values are defined as a probability of obtaining a joint empirical percentile greater than or equal to that observed for the window in question. For the joint empirical percentile, we defined the top 1% windows as outlier windows. As with our primary approach described above, windows  $\leq$  200kb apart were collapsed into outlier regions. For this method, we ranked outlier regions by the maximum joint percentile.

#### **Dog-specific mutations**

For our analysis of the distribution of sites fixed between dogs and wolves, we first identified sites where the Basenji and Dingo were homozygous for the Boxer reference derived allele, and where the three wolves and golden jackal were fixed for an alternative (i.e. the ancestral) allele. We then reduced this set of candidate sites by only including sites where the dog derived allele was observed across the 12-breed genome sequences at a frequency  $\geq 0.75$ . We evaluated the functional consequences of dog-specific non-synonymous variants using Ensembl's Variant Effect Predictor (http://www.ensembl.org/info/docs/tools/vep/index.html).

#### **Enrichment analyses**

To detect functional enrichment within the genes intersecting our outlier regions, we used the program DAVID [74], with the *Canis lupus* gene set as background. We focused on the genes falling within 25kb of the peak  $CMS_{1-FDR}$  signal for the top 100 regions, minus those regions that did not also show a reduction in diversity in the 12-breed data set. Because enrichment analyses require a relatively large input set of genes in order to detect enrichment patterns, and given that we already perform statistical inference to identify regions under selection, we report all categories with FDR  $\leq$  10%. We also report uncorrected P-values as well as P-values corrected for multiple comparisons using Benjamini's method, although the latter are generally considered to be extremely conservative [74].

### **Supporting Information**

**S1 Text. Description of additional candidate genes.** (PDF)

S1 Table. Outlier regions identified with the FDR-based method, ranked according to CMS<sub>1-FDR</sub>.

(XLS)

S1 Fig. Genome-wide distribution of  $\Delta \pi$ , F<sub>ST</sub>,  $\Delta$  Tajima's D in 100kb sliding windows. (PDF)

**S2 Fig.** Comparison of distributions computed from neutral coalescent simulations based up the posterior mean parameter estimates from the inferred demographic history, [19] and 1000 samples from the joint posterior distribution for (A)  $\Delta \pi$ , (B) F<sub>ST</sub>, and (C)  $\Delta$  Tajima's D. (PDF)

S3 Fig. Distribution of overlaps between outlier regions detected between methods for FDR and empirical outlier methods.

(PDF)

S4 Fig. Bi-plots of empirical percentile vs. FDR for individual summary statistics across 100 kb windows, demonstrating that the majority of windows in the top 1% have FDR > 0.01. (A) Entire range of empirical percentile and (B) Focus on the top 20% of the empirical distribution. Horizontal and vertical dotted lines indicate the 99<sup>th</sup> percentile and 1% FDR, respectively.

(PDF)

**S5 Fig.** FDR of individual statistics vs. the joint percentile statistic for 100kb windows, used to identify outlier windows in the empirical outlier (non-FDR) approach, for (A)  $\Delta\pi$ , (B) F<sub>ST</sub>, and (C)  $\Delta$  Tajima's D. (PNG)

**S6 Fig. Venn diagram displaying overlap of candidate gene sets obtained with FDR-based and empirical outlier (EO) methods for detecting positive selection on the dog lineage.** Genes unique to empirical methods relative to FDR methods are those falling within windows with a high false discovery rate (and thus are likely to be enriched with false positives) (PDF)

S7 Fig. Distribution of  $CMS_{1-FDR}$  and the joint percentile statistic for the top and 3<sup>rd</sup> ranked regions, demonstrating that  $CMS_{1-FDR}$  localizes the peak of the outlier region signal more precisely than the joint percentile. (PDF)

S8 Fig. All 68 outlier regions identified using the FDR-based methodology using  $\Delta \pi$ ,  $F_{ST}$ ,  $\Delta$  Tajima's D that were validated with the 12-breed dog diversity panel. Columns within "This study" are based on the sequencing data generated here, while those under CanMap are computed from a ~48k SNP data set for a large set of wolves and ancient/basal dog breeds. Heat map colors reflect upper percentiles of the calculated metrics, with warmer colors indicating higher percentiles. Overlaps with previous studies: 1, vonHoldt et al. 2010 [35]; 2, Vaysse et al. (2011) [25]; 3, Boyko et al. (2010) [23]; and Axelsson et al.(2013), [27], with numbers indicating the joint percentile,  $F_{ST}$ ,  $F_{ST}$ , and region id, respectively for each study. (PDF)

**S9** Fig. Distribution of sites fixed between dogs and wolves in neutral and outlier regions according to functional class, filtered according to the requirement that the dog-specific allele be at a frequency of 0.75 or greater among a panel of 12 additional breed dogs. Numbers above bars indicate counts of fixed sites. (PDF)

#### Acknowledgments

We thank B. Chin, T. Toy, Z. Chen and the UCLA DNA Microarray Facility for library preparations and sequencing done at UCLA; D. Wegmann for initial development of the *VcfAnnotator* program used in analyses here; A. Platt for feedback on analyses and manuscript; R. Hefner and The National Collections of Natural History at Tel Aviv University for procuring and access to samples. We thank E. Randi, R. Godinho, and B. Yue for facilitating visits of MG, PMS, and ZF to the lab of RKW. The computations in this paper were run on the Hoffmann2 cluster supported by the IDRE Research Technology Group at the University of California, Los Angeles, and the Odyssey cluster supported by the FAS Division of Science, Research Computing Group at Harvard University.

#### **Author Contributions**

Conceived and designed the experiments: AHF CDB TTH SFN EAO TMB RKW JN. Performed the experiments: RMS BLG OR CV KS ARB HGP CL VT. Analyzed the data: AHF RMS IG EH BWD DODV PMS MG ZF PM BLG OR FH CA VT AS JN. Contributed reagents/ materials/analysis tools: CV EG JK EAO RKW IG. Wrote the paper: AHF JN RKW.

#### References

- 1. Smith JM, Haigh J (1974) The hitch-hiking effect of a favourable gene. Genet Res 89: 391–403.
- Nielsen R, Williamson S, Kim Y, Hubisz MJ, Clark AG, et al. (2005) Genomic scans for selective sweeps using SNP data. Genome Res 15: 1566–1575. PMID: <u>16251466</u>
- Pavlidis P, Zivkovic D, Stamatakis A, Alachiotis N (2013) SweeD: Likelihood-Based Detection of Selective Sweeps in Thousands of Genomes. Mol Biol Evol 30: 2224–2234. doi: <u>10.1093/molbev/mst112</u> PMID: <u>23777627</u>
- 4. Voight BF, Kudaravalli S, Wen XQ, Pritchard JK (2006) A map of recent positive selection in the human genome. PLoS Biol 4: 446–458.
- 5. Barton NH (1998) The effect of hitch-hiking on neutral genealogies. Genet Res 72: 123–133.
- 6. Barton NH (2000) Genetic hitchhiking. Philos Trans R Soc London Ser B 355: 1553–1562.
- Pavlidis P, Jensen JD, Stephan W (2010) Searching for Footprints of Positive Selection in Whole-Genome SNP Data From Nonequilibrium Populations. Genetics 185: 907–922. doi: <u>10.1534/genetics.</u> <u>110.116459</u> PMID: <u>20407129</u>
- Slatkin M, Wiehe T (1998) Genetic hitch-hiking in a subdivided population. Genet Res 71: 155–160. PMID: <u>9717437</u>
- Santiago E, Caballero A (2005) Variation after a selective sweep in a subdivided population. Genetics 169: 475–483. PMID: <u>15489530</u>
- Jensen JD, Kim Y, DuMont VB, Aquadro CF, Bustamante CD (2005) Distinguishing between selective sweeps and demography using DNA polymorphism data. Genetics 170: 1401–1410. PMID: <u>15911584</u>
- Thornton KR, Jensen JD (2007) Controlling the false-positive rate in multilocus genome scans for selection. Genetics 175: 737–750. PMID: <u>17110489</u>
- Crisci JL, Poh YP, Mahajan S, Jensen JD (2013) The impact of equilibrium assumptions on tests of selection. Front Genet 4: 235. doi: <u>10.3389/fgene.2013.00235</u> PMID: <u>24273554</u>
- Akey JM, Ruhe AL, Akey DT, Wong AK, Connelly CF, et al. (2010) Tracking footprints of artificial selection in the dog genome. Proc Natl Acad Sci USA 107: 1160–1165. doi: <u>10.1073/pnas.0909918107</u> PMID: <u>20080661</u>

- Zeder MA (2015) Core questions in domestication research. Proc Natl Acad Sci USA 112: 3191–3198. doi: 10.1073/pnas.1501711112 PMID: 25713127
- Ovodov ND, Crockford SJ, Kuzmin YV, Higham TFG, Hodgins GWL, et al. (2011) A 33,000-Year-Old Incipient Dog from the Altai Mountains of Siberia: Evidence of the Earliest Domestication Disrupted by the Last Glacial Maximum. PLoS ONE 6.
- Germonpre M, Sablin MV, Stevens RE, Hedges REM, Hofreiter M, et al. (2009) Fossil dogs and wolves from Palaeolithic sites in Belgium, the Ukraine and Russia: osteometry, ancient DNA and stable isotopes. J Archaeol Sci 36: 473–490.
- Germonpre M, Laznickova-Galetova M, Sablin MV (2012) Palaeolithic dog skulls at the Gravettian Predmosti site, the Czech Republic. J Archaeol Sci 39: 184–202.
- Larson G, Karlsson EK, Perri A, Webster MT, Ho SYW, et al. (2012) Rethinking dog domestication by integrating genetics, archeology, and biogeography. Proc Natl Acad Sci USA 109: 8878–8883. doi: <u>10.</u> <u>1073/pnas.1203005109</u> PMID: <u>22615366</u>
- Freedman AH, Gronau I, Schweizer RM, Ortega-Del Vecchyo D, Han EJ, et al. (2014) Genome Sequencing Highlights the Dynamic Early History of Dogs. PLoS Genet 10:e1004016. doi: <u>10.1371/</u> journal.pgen.1004016 PMID: <u>24453982</u>
- 20. Thalmann O, Shapiro B, Cui P, Schuenemann VJ, Sawyer SK, et al. (2013) Complete Mitochondrial Genomes of Ancient Canids Suggest a European Origin of Domestic Dogs. Science 342: 871–874. doi: 10.1126/science.1243650 PMID: 24233726
- 21. Larson G, Fuller DQ (2014) The evolution of animal domestication Annu Rev Ecol Syst 45: 115–136.
- Saetre P, Lindberg J, Leonard JA, Olsson K, Pettersson U, et al. (2004) From wild wolf to domestic dog: gene expression changes in the brain. Mol Brain Res 126: 198–206. PMID: <u>15249144</u>
- Boyko AR, Quignon P, Li L, Schoenebeck JJ, Degenhardt JD, et al. (2010) A Simple Genetic Architecture Underlies Morphological Variation in Dogs. Plos Biol 8:e1000451. doi: <u>10.1371/journal.pbio.</u> 1000451 PMID: 20711490
- Cadieu E, Neff MW, Quignon P, Walsh K, Chase K, et al. (2009) Coat Variation in the Domestic Dog Is Governed by Variants in Three Genes. Science 326: 150–153. doi: <u>10.1126/science.1177808</u> PMID: 19713490
- Vaysse A, Ratnakumar A, Derrien T, Axelsson E, Pielberg GR, et al. (2011) Identification of Genomic Regions Associated with Phenotypic Variation between Dog Breeds using Selection Mapping. Plos Genet 7: e1005059.
- Karlsson EK, Baranowska I, Wade CM, Salmon Hillbertz NHC, Zody MC, et al. (2007) Efficient mapping of mendelian traits in dogs through genome-wide association. Nat Genet 39: 1321–1328. PMID: <u>17906626</u>
- Axelsson E, Ratnakumar A, Arendt ML, Maqbool K, Webster MT, et al. (2013) The genomic signature of dog domestication reveals adaptation to a starch-rich diet. Nature 495: 360–364. doi: <u>10.1038/</u> nature11837 PMID: 23354050
- 28. Wang G-D, Zhai WW, Yang H-C, Fan R-X, Cao X, et al. (2013) The genomics of selection in dogs and the parallel evolution between dogs and humans. Nat Comm 4:1860.
- 29. Gray MM, Granka JM, Bustamante CD, Sutter NB, Boyko AR, et al. (2009) Linkage Disequilibrium and Demographic History of Wild and Domestic Canids. Genetics 181: 1493–1505. doi: <u>10.1534/genetics.</u> <u>108.098830</u> PMID: <u>19189949</u>
- Lindblad-Toh K, Wade CM, Mikkelsen TS, Karlsson EK, Jaffe DB, et al. (2005) Genome sequence, comparative analysis and haplotype structure of the domestic dog. Nature 438: 803–819. PMID: <u>16341006</u>
- Anderson TM, vonHoldt BM, Candille SI, Musiani M, Greco C, et al. (2009) Molecular and Evolutionary History of Melanism in North American Gray Wolves. Science 323: 1339–1343. doi: <u>10.1126/science</u>. 1165448 PMID: 19197024
- vonHoldt BM, Pollinger JP, Earl DA, Knowles JC, Boyko AR, et al. (2011) A genome-wide perspective on the evolutionary history of enigmatic wolf-like canids. Genome Res 21: 1294–1305. doi: <u>10.1101/gr.</u> <u>116301.110</u> PMID: <u>21566151</u>
- Larson G, Burger J (2013) A population genetics view of animal domestication. Trends Genet 29: 197– 205. doi: <u>10.1016/j.tig.2013.01.003</u> PMID: <u>23415592</u>
- Vila C, Seddon J, Ellegren H (2005) Genes of domestic mammals augmented by backcrossing with wild ancestors. Trends Genet 21: 214–218. PMID: <u>15797616</u>
- vonHoldt BM, Pollinger JP, Lohmueller KE, Han EJ, Parker HG, et al. (2010) Genome-wide SNP and haplotype analyses reveal a rich history underlying dog domestication. Nature 464: 898–U109. doi: <u>10.</u> <u>1038/nature08837</u> PMID: <u>20237475</u>

- Qanbari S, Pausch H, Jansen S, Somel M, Strom TM, et al. (2014) Classic Selective Sweeps Revealed by Massive Sequencing in Cattle. PLoS Genet 10:e1004148. doi: <u>10.1371/journal.pgen.1004148</u> PMID: <u>24586189</u>
- Li M, Tian S, Jin L, Zhou G, Li Y, et al. (2013) Genomic analyses identify distinct patterns of selection in domesticated pigs and Tibetan wild boars. Nat Genet 45: 1431–1438. doi: <u>10.1038/ng.2811</u> PMID: <u>24162736</u>
- Miller W, Schuster SC, Welch AJ, Ratan A, Bedoya-Reina OC, et al. (2012) Polar and brown bear genomes reveal ancient admixture and demographic footprints of past climate change. Proc Natl Acad Sci USA 109: E2382–E2390. doi: <u>10.1073/pnas.1210506109</u> PMID: <u>22826254</u>
- Skoglund P, Ersmark E, Palkopoulou E, Dalen L (2015) Ancient wolf genome reveals an early divergence of domestic dog ancestors and admixture into high-latitude breeds. Curr Biol 25: 1515–1519. doi: 10.1016/j.cub.2015.04.019 PMID: 26004765
- 40. Innan H, Kim Y (2008) Detecting local adaptation using the joint sampling of polymorphism data in the parental and derived populations. Genetics 179: 1713–1720. doi: <u>10.1534/genetics.108.086835</u> PMID: 18562650
- Storey JD, Tibshirani R (2003) Statistical significance for genomewide studies. Proc Natl Acad Sci USA 100: 9440–9445. PMID: <u>12883005</u>
- Benjamini Y, Hochberg Y (1995) Controlling the False Discovery Rate—a Practical and Powerful Approach to Multiple Testing. J R Stat Soc Ser B Met 57: 289–300.
- Grossman SR, Shylakhter I, Karlsson EK, Byrne EH, Morales S, et al. (2010) A Composite of Multiple Signals Distinguishes Causal Variants in Regions of Positive Selection. Science 327: 883–886. doi: 10.1126/science.1183863 PMID: 20056855
- Trut L, Oskina I, Kharlamova A (2009) Animal evolution during domestication: the domesticated fox as a model. Bioessays 31: 349–360. doi: 10.1002/bies.200800070 PMID: 19260016
- Milinkovic V, Bankovic J, Rakic M, Stankovic T, Skender-Gazibara M, et al. (2013) Identification of novel genetic alterations in samples of malignant glioma patients. PLoS One 8: e82108. doi: <u>10.1371/</u> journal.pone.0082108 PMID: 24358143
- Maestrini E, Pagnamenta AT, Lamb JA, Bacchelli E, Sykes NH, et al. (2010) High-density SNP association study and copy number variation analysis of the AUTS1 and AUTS5 loci implicate the IMMP2L-DOCK4 gene region in autism susceptibility. Mol Psychiatry 15: 954–968. doi: <u>10.1038/mp.2009.34</u> PMID: 19401682
- Casey JP, Magalhaes T, Conroy JM, Regan R, Shah N, et al. (2012) A novel approach of homozygous haplotype sharing identifies candidate genes in autism spectrum disorder. Hum Genet 131: 565–579. doi: <u>10.1007/s00439-011-1094-6</u> PMID: <u>21996756</u>
- Minelli A, Scassellati C, Bonvicini C, Perez J, Gennarelli M (2009) An Association of GRIK3 Ser310Ala Functional Polymorphism with Personality Traits. Neuropsychobiology 59: 28–33. doi: <u>10.1159/</u> 000202827 PMID: 19221446
- 49. Cherlyn SYT, Woon PS, Liu JJ, Ong WY, Tsai GC, et al. (2010) Genetic association studies of glutamate, GABA and related genes in schizophrenia and bipolar disorder: A decade of advance. Neurosci Biobehav Rev 34: 958–977. doi: 10.1016/j.neubiorev.2010.01.002 PMID: 20060416
- Schmidt A, Wolde M, Thiele C, Fest W, Kratzin H, et al. (1999) Endophilin I mediates synaptic vesicle formation by transfer of arachidonate to lysophosphatidic acid. Nature 401: 133–141. PMID: <u>10490020</u>
- Ayalew M, Le-Niculescu H, Levey DF, Jain N, Changala B, et al. (2012) Convergent functional genomics of schizophrenia: from comprehensive understanding to genetic risk prediction. Molecular Psychiatry 17: 887–905. doi: <u>10.1038/mp.2012.37</u> PMID: <u>22584867</u>
- de Gortari P, Mengod G (2010) Dopamine D1, D2 and mu-opioid receptors are co-expressed with adenylyl cyclase 5 and phosphodiesterase 7B mRNAs in striatal rat cells. Brain Res 1310: 37–45. doi: <u>10.</u> <u>1016/j.brainres.2009.11.009</u> PMID: <u>19913519</u>
- Kwon YT, Balogh SA, Davydov IV, Kashina AS, Yoon JK, et al. (2000) Altered activity, social behavior, and spatial memory in mice lacking the NTAN1p amidase and the asparagine branch of the N-end rule pathway. Mol Cell Biol 20: 4135–4148. PMID: <u>10805755</u>
- Balogh SA, McDowell CS, Kwon YT, Denenberg VH (2001) Facilitated stimulus-response associative learning and long-term memory in mice lacking the NTAN1 amidase of the N-end rule pathway. Brain Res 892: 336–343. PMID: <u>11172781</u>
- Durinck S, Stawiski EW, Pavia-Jimenez A, Modrusan Z, Kapur P, et al. (2015) Spectrum of diverse genomic alterations define non-clear cell renal carcinoma. Nat Genet 47: 13–21. doi: <u>10.1038/ng.3146</u> PMID: <u>25401301</u>
- 56. Haraksingh RR, Jahanbani F, Rodriguez-Paris J, Gelernter J, Nadeau KC, et al. (2014) Exome sequencing and genome-wide copy number variant mapping reveal novel associations with

sensorineural hereditary hearing loss. Bmc Genomics 15: 1155. doi: <u>10.1186/1471-2164-15-1155</u> PMID: <u>25528277</u>

- Shiang R, Ryan SG, Zhu YZ, Hahn AF, Oconnell P, et al. (1993) Mutations in the Alpha-1 Subunit of the Inhibitory Glycine Receptor Cause the Dominant Neurologic Disorder, Hyperekplexia. Nat Genet 5: 351–358. PMID: 8298642
- Kawai M, Rosen CJ (2010) PPAR gamma: a circadian transcription factor in adipogenesis and osteogenesis. Nat Rev Endocr 6: 629–636.
- Green CB, Douris N, Kojima S, Strayer CA, Fogerty J, et al. (2007) Loss of Nocturnin, a circadian deadenylase, confers resistance to hepatic steatosis and diet-induced obesity. Proc Natl Acad Sci USA 104: 9888–9893. PMID: <u>17517647</u>
- 60. Kawai M, Green CB, Lecka-Czernik B, Douris N, Gilbert MR, et al. (2010) A circadian-regulated gene, Nocturnin, promotes adipogenesis by stimulating PPAR-gamma nuclear translocation. Proc Natl Acad Sci USA 107: 10508–10513. doi: 10.1073/pnas.1000788107 PMID: 20498072
- Kawai M, Delany AM, Green CB, Adamo ML, Rosen CJ (2010) Nocturnin Suppresses Igf1 Expression in Bone by Targeting the 3 ' Untranslated Region of Igf1 mRNA. Endocrinology 151: 4861–4870. doi: 10.1210/en.2010-0407 PMID: 20685873
- Hoopes BC, Rimbault M, Liebers D, Ostrander EA, Sutter NB (2012) The insulin-like growth factor 1 receptor (IGF1R) contributes to reduced size in dogs. Mamm Genome 23: 780–790. doi: <u>10.1007/</u> s00335-012-9417-z PMID: 22903739
- Sutter NB, Bustamante CD, Chase K, Gray MM, Zhao KY, et al. (2007) A single IGF1 allele is a major determinant of small size in dogs. Science 316: 112–115. PMID: 17412960
- Fuchs M, Hafer A, Munch C, Kannenberg F, Teichmann S, et al. (2001) Disruption of the sterol carrier protein 2 gene in mice impairs biliary lipid and hepatic cholesterol metabolism. J Biol Chem 276: 48058–48065. PMID: 11673458
- Demirkan A, van Duijn CM, Ugocsai P, Isaacs A, Pramstaller PP, et al. (2012) Genome-Wide Association Study Identifies Novel Loci Associated with Circulating Phospho- and Sphingolipid Concentrations. PLoS Genet 8:e1002490. doi: <u>10.1371/journal.pgen.1002490</u> PMID: <u>22359512</u>
- Manceau M, Domingues VS, Mallarino R, Hoekstra HE (2011) The Developmental Role of Agouti in Color Pattern Evolution. Science 331: 1062–1065. doi: <u>10.1126/science.1200684</u> PMID: <u>21350176</u>
- **67.** Linnen CR, Poh YP, Peterson BK, Barrett RDH, Larson JG, et al. (2013) Adaptive Evolution of Multiple Traits Through Multiple Mutations at a Single Gene. Science 339: 1312–1316. doi: <u>10.1126/science.</u> <u>1233213 PMID: 23493712</u>
- Xue BZ, Moustaid-Moussa N, Wilkison WO, Zemel MB (1998) The agouti gene product inhibits lipolysis in human adipocytes via a Ca2+-dependent mechanism. FASEB J 12: 1391–1396. PMID: <u>9761782</u>
- Carola V, Perlas E, Zonfrillo F, Soini HA, Novotny MV, et al. (2014) Modulation of social behavior by the agouti pigmentation gene. Front Behav Neurosci 8:259. doi: <u>10.3389/fnbeh.2014.00259</u> PMID: 25136298
- Hayssen V (1997) Effects of the nonagouti coat-color allele on behavior of deer mice (Peromyscus maniculatus): a comparison with Norway rats (Rattus norvegicus). J Comp Psychol 111: 419–423. PMID: <u>9419886</u>
- Liu F, Wollstein A, Hysi PG, Ankra-Badu GA, Spector TD, et al. (2010) Digital Quantification of Human Eye Color Highlights Genetic Association of Three New Loci. PLoS Genet 6:e1000934. doi: <u>10.1371/</u> journal.pgen.1000934 PMID: <u>20463881</u>
- Jackson IJ (1997) Homologous pigmentation mutations in human, mouse and other model organisms. Hum Mol Genet 6: 1613–1624. PMID: <u>9300652</u>
- Pei Z, Jia Z, Watkins PA (2006) The second member of the human and murine bubblegum family is a testis- and brainstem-specific acyl-CoA synthetase. J Biol Chem 281: 6632–6641. PMID: <u>16371355</u>
- Huang DW, Sherman BT, Lempicki RA (2009) Systematic and integrative analysis of large gene lists using DAVID bioinformatics resources. Nature Protoc 4: 44–57.
- 75. Shipman P The invaders: how humans and their dogs drove Neanderthals to extinction. Cambridge, Massachusetts: Belknap Press.
- Shipman P (2015) How do you kill 86 mammoths? Taphonomic investigations of mammoth megasites. Quat Int 359: 38–46.
- Albert FW, Carlborg O, Plyusnina I, Besnier F, Hedwig D, et al. (2009) Genetic Architecture of Tameness in a Rat Model of Animal Domestication. Genetics 182: 541–554. doi: <u>10.1534/genetics.109.</u> <u>102186</u> PMID: <u>19363126</u>

- 78. Albert FW, Somel M, Carneiro M, Aximu-Petri A, Halbwax M, et al. (2012) A comparison of brain gene expression levels in domesticated and wild animals. PLoS Genet 8: e1002962. doi: <u>10.1371/journal.pgen.1002962</u> PMID: <u>23028369</u>
- 79. Montague MJ, Li G, Gandolfi B, Khan R, Aken BL, et al. (2014) Comparative analysis of the domestic cat genome reveals genetic signatures underlying feline biology and domestication. Proc Natl Acad Sci USA 111: 17230–17235. doi: <u>10.1073/pnas.1410083111</u> PMID: <u>25385592</u>
- Heyne HO, Lautenschlager S, Nelson R, Besnier F, Rotival M, et al. (2014) Genetic Influences on Brain Gene Expression in Rats Selected for Tameness and Aggression. Genetics 198: 1277–+. doi: <u>10.</u> <u>1534/genetics.114.168948</u> PMID: <u>25189874</u>
- Wilkins AS, Wrangham RW, Fitch WT (2014) The 'Domestication Syndrome' in Mammals: A Unified Explanation Based on Neural Crest Cell Behavior and Genetics (vol 197, pg 795, 2014). Genetics 198: 1771–1771.
- Teshima KM, Coop G, Przeworski M (2006) How reliable are empirical genomic scans for selective sweeps? Genome Res 16: 702–712. PMID: <u>16687733</u>
- Messer PW, Petrov DA (2013) Population genomics of rapid adaptation by soft selective sweeps. Trends Ecol Evol 28: 659–669. doi: 10.1016/j.tree.2013.08.003 PMID: 24075201
- 84. Jensen JD (2014) On the unfounded enthusiasm for soft selective sweeps. Nat Comm 5:5281.
- Benjamini Y, Yekutieli D (2001) The control of the false discovery rate in multiple testing under dependency. Ann Stat 29: 1165–1188.
- Sabatti C, Service S, Freimer N (2003) False discovery rate in linkage and association genome screens for complex disorders. Genetics 164: 829–833. PMID: <u>12807801</u>
- 87. Slatkin M, Wiehe T (1998) Genetic hitch-hiking in a subdivided population. Genetical Res 71: 155–160.
- Buckler ES, Thornsberry JM, Kresovich S (2001) Molecular diversity, structure and domestication of grasses. Genetical Res 77: 213–218.
- Innan H, Kim Y (2004) Pattern of polymorphism after strong artificial selection in a domestication event. Proc Natl Acad Sci USA 101: 10667–10672. PMID: <u>15249682</u>
- 90. Tajima F (1989) Statistical-Method for Testing the Neutral Mutation Hypothesis by DNA Polymorphism. Genetics 123: 585–595. PMID: <u>2513255</u>
- Hudson RR (2002) Generating samples under a Wright-Fisher neutral model of genetic variation. Bioinformatics 18: 337–338. PMID: <u>11847089</u>